Author's response to reviews

Title: Novel Unbalanced 16p;13q Translocation in a Patient with Factor VII Deficiency and Developmental Abnormalities: Case Report

Authors:

Brian P Brooks (brooksbl@mail.nih.gov)
Jeanne M Meck (meckj@gunet.georgetown.edu)
Bassem R Haddad (haddadb1@georgetown.edu)
Claude Bendavid (claude.bendavid@univ-rennes1.fr)
Delphine Blain (dblain@mail.nih.gov)
Jeffrey A Toretsky (jat42@georgetown.edu)

Version: 4 Date: 8 November 2005

Author's response to reviews: see over
Dear Editors:

We are pleased to submit our revised manuscript (MS: 2921013717014216), “Novel Unbalanced 16p:13q Translocation in a Patient with Factor VII Deficiency and Developmental Abnormalities: Case Report.” We have made all the requested changes.

The editors requested that we expand the background section of the abstract, which we have done. We have also provided the release to publish photos as a supplementary document.

Reviewer #1 had no specific comments for revision.

Reviewer #2 requested we correct the grammar in the abstract, correct the spelling of “cytogenetic” and add a “wrap-up” paragraph, which we have done.

Thank you for considering our manuscript. We look forward to hearing from you.

Sincerely,

Brian P. Brooks, M.D., Ph.D.
National Eye Institute & National Human Genome Research Institute